Author’s response to reviews

Title: Mucinous cystic neoplasm of the liver with extrahepatic growth presenting with ascending cholangitis diagnosed by EUS features: A case report

Authors:

Tanawat Pattarapuntakul (tanawat_kuey@hotmail.com)

Bancha Ovartlanporn (obancha@live.com)

Jaksin Sottisuporn (pondjaksin@hotmail.com)

Version: 2 Date: 20 Dec 2017

Author’s response to reviews:

I would like to thanks editors and reviewers, who correct and advise our team some point for correction this case report. I have been revised my manuscripts according to the recommendations.

1. I have been added references in introduction section.

2. I have been correct and revised my all manuscript with my native speaker of English. The patient’s information was rearranged the time line of investigations and management.

3. I have been added a unique point of this case report about the EUS features can helped us to the diagnosis in this case.

4. I have been revision list the same authors as first submission, I am so sorry that I am confuse to add another person in authorship, she has been joined with us in another paper.

Thank you

Pattarapuntakul T.